Author's response to reviews

Title: Delay in renal replacement therapy initiation increases mortality in patients developing acute kidney injury after liver transplantation

Authors:

Roberto C Narciso (rnarciso@gmail.com)
Leonardo R Ferraz (p.polimanti@uol.com.br)
Sergio Mies (smies@usp.br)
Julio CM Monte (icesarmonte@nefro.epm)
Oscar FP dos Santos (pavao@nefro.epm.br)
Miguel Cendoroglo (cendoro@uol.com.br)
Cassio JO Rodrigues (cjrrodrigues@uol.com.br)
Marcelo C Batista (marcelocbatista@uol.com.br)
Marcelino S Durão Jr (m.durao@uol.com.br)

Version: 2 Date: 28 September 2012

Author's response to reviews:

We would like to thank the evaluation of our manuscript entitled: referenced above. All reviewer’s comments were considered and the manuscript revised. The issues raised by the reviewers were carefully answered and the necessary modifications fulfilled on the body text.

Thank you very much,

With kind regards